DOI: 10.1002/agj2.70098

ORIGINAL ARTICLE

Agronomy, Soils, and Environmental Quality

Agronomy Journal

Annual ribwort plantain and alfalfa mixtures enhance forage accumulation and reduce nitrate

Abstract

Tsvetelina Krachunova^{1,2,3}

Sonoko Bellingrath-Kimura^{2,3} Knut Schmidtke¹

¹Faculty of Agriculture/Environment/ Chemistry, Professorship of Organic farming Dresden University of Applied Science, Dresden, Germany

²Faculty of Life Sciences, Thaer-Institute of Agricultural and Horticultural Science, Humboldt University of Berlin, Berlin, Germany

³Leibniz Centre for Agricultural Landscape Research (ZALF), Müncheberg, Germany

Correspondence

Tsvetelina Krachunova, Leibniz Centre for Agricultural Landscape Research (ZALF), Eberswalder Straße 84, 15374, Müncheberg, Germany.

Email: Tsvetelina.Krachunova@zalf.de

Assigned to Associate Editor Lisa Baxter.

Funding information

eip-agri Agriculture & Innovation-Rural development 2014-2020 for Operational Groups (in the sense of EU Art 56 of Reg. 1305/2013), Grant/Award Number: 332018001101LWC; Dresden University of Applied Sciences

Nitrate (NO₃) leaching from alfalfa (Medicago sativa L.) cultivation in autumn and sustaining high forage accumulation under dry conditions is a serious problem in farming. In this study, mixtures of alfalfa and ribwort plantain (Plantago lanceolata L.) had agronomic advantages for forage accumulation compared to corresponding mixtures of alfalfa and a grass species (meadow fescue [Festuca pratensis Hudson]). The ribwort plantain and alfalfa mixtures accumulated twice as much forage as the reference mixtures with meadow fescue. Most of the forage accumulation was accounted for ribwort plantain due to poor initial alfalfa development. Ribwort plantain suppressed alfalfa and all weed species from as early as July and continued increasing forage accumulation until early autumn. At the same time, ribwort plantain contributed more than meadow fescue to a reduction in NO₃-N. Significantly lower NO₃-N shares in the soil were observed, on average, in mixtures with ribwort plantain and alfalfa in the 0.6- to 1.2-m soil depths in autumn. Due to the inhibition of nitrification by ribwort plantain, NH₄-N was present in the soil solution of the sandy soil at the trial site to a greater extent in autumn compared to alfalfa and the mixtures with meadow fescue. Due to the highly competitive power of ribwort plan-

Plain Language Summary

seeds m^{-2} .

Nitrate (NO₃) leaching from alfalfa (Medicago sativa L.) cultivation in autumn and sustaining high forage accumulation under dry conditions is a serious problem in farming. In this study, mixtures of alfalfa and ribwort plantain (Plantago lanceolata L.) had agronomic advantages for forage accumulation compared to corresponding mixtures of alfalfa and a grass species (meadow fescue [Festuca pratensis Hudson]). The ribwort plantain and alfalfa mixtures accumulated twice as much forage as the reference mixtures with meadow fescue. Most of the forage accumulation

tain, it should only be sown in mixtures with alfalfa at seed rates of 100 germinating

Abbreviations: LER, land equivalent ratio; pLER, partial land equivalent ratio.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

© 2025 The Author(s). Agronomy Journal published by Wiley Periodicals LLC on behalf of American Society of Agronomy.

was accounted for ribwort plantain due to poor initial alfalfa development. Ribwort plantain suppressed alfalfa and all weed species from as early as July and continued increasing forage accumulation until early autumn. At the same time, ribwort plantain contributed more than meadow fescue to a reduction in NO₃-N. Significantly lower NO₃-N shares in the soil were observed, on average, in mixtures with ribwort plantain and alfalfa.

1 | INTRODUCTION

The goal of organic farming is to develop environmentally sustainable and integrated crop production systems that work similarly to natural ecosystems (Rigby & Cáceres, 2001). Organic cultivation methods completely exclude the use of synthetic nitrogen (N) fertilizers (Hansen et al., 2000; Knapp et al., 2023), thus organic farming practices must provide a sufficient N supply through symbiotic N₂ fixation to maintain a moderate yield level (Barbieri et al., 2021). In organic farming, N loss in a certain crop growth cycle leads to higher yield losses compared to conventional farming practices, which can compensate N losses easier through mineral fertilizers (Pandey et al., 2018).

Due to the ability of fodder legumes to symbiotically fix N, they are a necessary component of crop rotations in organic farming (Fustec et al., 2010; Kayser et al., 2010). Alfalfa (Medicago sativa L.) is a high-yielding, deep-rooted, and crude protein-rich fodder legume capable of symbiotically fixing high amounts of N; it is also characterized by high drought tolerance (Ghimire et al., 2021; Moghaddam et al., 2015). Alfalfa is capable of achieving higher yields than other fodder legumes (e.g., clovers) on organic farms (Liu et al., 2022). However, alfalfa cultivation can produce very high quantities of NO₃-N that are accumulated as a result of the input of high and easily mineralizable organically bound N quantities (Heichel et al., 1984). Alfalfa can fix between 93 and 183 kg N ha⁻¹, on average, in the first year (Burity et al., 1989; Kelner et al., 1997). Rakotovololona et al. (2019) reported an average of 333 kg N ha⁻¹ fixed by alfalfa in pure stands. Numerous studies have shown that NO₃-N leaching in pure alfalfa stands is comparatively high in autumn of the first year (Burity et al., 1989; Heichel et al., 1984, 1985; Masoni et al., 2015; Rakotovololona et al., 2019; Schmidtke, 2001; Ta & Faris, 1987; Walley et al., 1996). This creates a concern for organic farming as a means by which loss of critical resources to maintain yield and retain N in the system may occur. NO₃-N leaching from perennially grown alfalfa is potentially lower (Benoit et al., 2014), which is associated with a high N content in harvested biomass and a greater water depletion through the roots of alfalfa (Singh et al., 2023).

Alfalfa is often mixed with grass species, which is beneficial for farming systems (Fernandez et al., 2019). In many cases, alfalfa and grass mixtures achieve higher yields compared to pure stands (Aponte et al., 2019; Fernandez et al., 2019; Sleugh et al., 2000; Veronesi et al., 2010). Meadow fescue (Festuca pratensis Hudson) is a valuable grass for alfalfa mixtures because it has a rather low competitive ability due to its slower establishment (Cherney et al., 2020; Flynn et al., 2013; Suter et al., 2004). Meadow fescue has a good fodder value, comparable to English ryegrass (Lolium perenne L.) (Suter et al., 2011). It is mainly found in lowto medium-intensity managed natural pastures, but it is also suitable for intensively managed meadows because of its high fodder value (Frick et al., 2019). Alfalfa and meadow fescue grow well together as an annual mixture because they both have similar growth seasons, drought tolerance, and a high fodder value (Hartmann, 2013; Kivelitz, 2020; Landwirtschaftskammer NRW, 2025; LfL, 2025).

Ribwort plantain (Plantago lanceolata L.) is a widely distributed plant species in meadows and pastures across Central Europe (Pol et al., 2021) and is palatable for grazing animals (Cavers et al., 1980; Rumball et al., 1997; Stewart, 1996). Most frequently, ribwort plantain in agriculture is as a component of mixed pastures (Stewart, 1996). However, ribwort plantain offers agronomic properties that have received little recognition in agriculture but have potential to increase the resilience and yield of plant stands in organic farming. Ribwort plantain has an average root length density of 1.6 cm^{-3} in 0- to 0.75-m soil depth and a specific root length up to 1.9 m, which can increase the resilience of cropping systems during drought periods under climate change (Pol et al., 2021). Ribwort plantain seeds germinate well under field conditions (Sagar & Harper, 1960), are insensitive to light conditions (Ellenberg & Leuschner, 2010; Thompson & Grime, 1983) and temperature changes (Thompson & Grime, 1983), and germination temperatures range from 13.5 to 30.5°C (Grey et al., 2019). Ribwort plantain is an encroaching species that can also be established through dormant seeding in winter (Grey et al., 2019; Pons & van der Toorn, 1988).

Furthermore, ribwort plantain contains aucubin, an iridoid glycoside that inhibits the nitrification of NH_4 -N and consequently reduces the (NO₃-N) content in the soil (Dietz et al., 2013). The secondary components acteoside (phenyl-propanoid) and catalpol also contribute to the inhibition of nitrification (Carlton et al., 2019). Moreover, ribwort plantain

3 of 20

reduces nitrous oxide gas (N₂O) formation in grasslands on peat soil by 39% compared to ryegrass (Pijlman et al., 2020) and by 50% in pastures (Gardiner et al., 2018) and reduces NO₃-N leaching by 19% in grazing systems (Navarrete et al., 2022).

Previously developed agronomic strategies to reduce NO₃-N leaching after the plowing of alfalfa in autumn, such as the cultivation of catch crops or main crops with high N uptake, postponement of the plowing date, and reduction of soil cultivation intensity, reduced nitrate (NO₃) losses after the plowing of alfalfa in winter (Dreymann et al., 2005; Heß, 1989; Koch, 1997; Notaris et al., 2018) but did not lead to a sustainable solution for the problem in organic farming. As a result, a novel climate-resilient cultivation strategy for alfalfa plowing in autumn is needed to generate high yields under climate change, to reduce NO₃ leaching in winter, and to diversify crop rotations in organic farming.

Despite the putative benefits of ribwort plantain, research regarding viability and sustainability in cultivation with alfalfa has not been done. Therefore, the objective of this study was to compare pure annual stands and different mixtures of alfalfa with ribwort plantain and meadow fescue under organic conditions for the first time. We hypothesized that (a) a mixture of alfalfa and ribwort plantain would result in a faster decline in the NO₃-N content in the soil layer compared to the reference mixture with meadow fescue by increasing the proportion of ammonium nitrogen (NH₄-N); and (b) a mixture of alfalfa with ribwort plantain would be higher yielding than a mixture of alfalfa with meadow fescue under increasing water shortage during the growing season.

2 **MATERIALS AND METHODS**

2.1 **Experimental site and design**

Different mixtures of ribwort plantain, alfalfa, and meadow fescue and pure stands were established at the same location in Canitz, Germany (near Leipzig, Germany) during 2019 and 2021. The site is located 112 m above sea level at 51°24'45" N 12°41'10" E. The field trial was in a water protection area. According to the Federal Water Resources Act (Bundesministerium der Justiz und für Verbraucherschutz, 2009/22.12.2023), the Fertilizer Regulation in Germany (DüV, 2017/11.12.2024) and the European Union Nitrate Directive (European Union, 1991), the upper limit for the application of N is 170 kg ha⁻¹ during specific periods on arable land in water protection areas. If the NO₃⁻ content in the groundwater exceeds 50 mg L^{-1} , any fertilization with liquid manure, slurry, compost, or solid manure is strictly prohibited. Previous site management and winter crop management are described in Table 1. The Canitz field has been farmed organically with this crop rotation since 1990 (Table 1).

Core Ideas

- Alfalfa in pure stands and in grass mixtures leads to nitrate losses in autumn.
- · Plant-available nitrogen was reduced under alfalfa with ribwort plantain in autumn.
- · Ribwort plantain shifted nitrate nitrogen to ammonium in alfalfa mixtures in autumn.
- · Mixture of ribwort plantain accumulated twice as much forage as meadow fescue mixtures.
- · Meadow fescue had a lower partial LER than ribwort plantain.

TABLE	1	Crop rotation for both study trials on the site Canitz for
the period	2016	-2021.

Сгор	Year	Crop rotation
Alfalfa (Medicago sativa L.)	2016-2017	Main crop
Spelt (Triticum spelta L.)	2018	Main crop
Phacelia (Phacelia tanacetifolia L.)	2018-2019	Catch crop
Field pea (Pisum sativum L.)	2019	Main crop
Oil radish (Raphanus sativus L.)	2019-2020	Catch crop
Potatoes (Solanum tuberosum L.)	2020	Main crop
Phacelia (Phacelia tanacetifolia L.)	2020-2021	Catch crop

The plots were arranged in a randomized block design with four replicates and random treatment allocations in each trial year (Figure 1A). The composition of ribwort plantain, alfalfa, and meadow fescue comprised a de Witt replacement series design from 0% to 100% of the respective plant species (alfalfa, ribwort plantain, and meadow fescue) to cultivate plant species in pure stands (100%) and pair them with a mixture of species in increasing percentages, as described by deWit (1960). Therefore, the following forage species mixtures were tested: (1) 100% ribwort plantain; (2) 67% ribwort plantain and 33% alfalfa; (3) 50% ribwort plantain and 50% alfalfa; (4) 33% ribwort plantain and 67% alfalfa; (5) 100% alfalfa; (6) 67% alfalfa and 33% meadow fescue; (7) 50% alfalfa and 50% meadow fescue; (8) 33% alfalfa and 67% meadow fescue; and (9) 100% meadow fescue. A plot was comprised of four drill passes for a total width of 7.2 m in width and 10 m in length (Figure 1A). The total experimental field dimensions were 85.2 m \times 55 m. The field trials were set up on April 1, 2019, and March 22, 2021.

All measurements and samplings described in the following sections took place within the second and third drill passes of the plots. The first and fourth drill passes minimized the border effects of the neighboring plots.

The 2 years of study were characterized by very different weather conditions. The warmest year was 2019 with an average temperature of 12.5°C and trial year 2021 was



FIGURE 1 (A) Experimental design as a randomized block ($85.2 \text{ m} \times 55 \text{ m}$) with each plot ($7.2 \text{ m} \times 10 \text{ m}$) consisting of four drill passes. (B) Mean monthly temperature and long-term temperature mean (°C) and (C) mean monthly precipitation and long-term mean of precipitation amount (mm) graphs for field trials at Canitz, Germany, from March to September 2019 and 2021.Weather data *source*: Wassergut Canitz GmbH.

significantly cooler at 9.8° C (Figure 1B). In March 2019, the average monthly temperature was almost 2°C above the long-term average. In April 2019, the average temperature remained in line with the long-term value, while April 2021 was 6.3° C below the long-term average at 9.9° C.

The trial site was characterized by low precipitation sums, averaging 557.3 mm per year and 370.7 mm for the March–September period (long-term means for the period 1913–2020), with total precipitation values of 253.4 mm in 2019 and 374.6 mm in 2021 (Figure 1C). May and August 2021 were exceptions, as there was 85.5% and 65.9% mm more precipitation compared to the long-term average for the period 1913–2020. The weather data were provided by the weather station at the Wassergut Canitz GmbH, farm in cooperation with Leipziger Stadtwerke.

Figure S1 captures the effect of climate change on daily temperature and precipitation values at the trial site from 2009 to 2023.

2.2 | Soil characterization and sampling

The site is characterized by Luvisols (para-brown soil) of periglacial gravel, with leading clay over deep sandy gravel.

The soil texture is loamy sand (IS) (LfULG, 2020). The average pH value was 6.8, which was determined following the method described by VDLUFA (2012c). The available water capacity ranged from 121 to 180 mm, and the field capacity ranged from 181 to 270 mm. The soil compaction sensitivity level was medium (LfULG, 2020). The experimental field also showed a high sensitivity to water and a medium sensitivity to wind erosion (LfULG, 2020).

When the field trials were set up (April 1, 2019, and March 22, 2021), soil material was collected to analyze the plant-available phosphorus (P), plant-available potassium (K), and plant-available magnesium (Mg) contents at a depth of 0-0.3 m using a Pürckhauer soil auger (diameter 0.03 m, sampling length: 0.3 m). Sampling points were randomly selected throughout the trial site according to the recommendations for soil characterization of Lorenz and Erdle (2018). Plant-available in this study refers to the fraction of total P, K, and Mg contents in soil available for uptake by plant roots. The P and K contents were determined following the method described by VDLUFA (2012b) and amounted to 6.5 and 14.9 mg 100 g soil⁻¹, respectively, in 0- to 0.3-m soil depth. The Mg content was determined following the method of Walinga et al. (1995) and was 11.8 mg 100 g soil⁻¹, on average.

The target N content in soil for farms in water protection areas in Germany is 30 kg N ha⁻¹ in order to reduce the nitrate levels in groundwater (Bundesministerium der Justiz und für Verbraucherschutz, 2009/22.12.2023). In the last 15 years, the farm has recorded an average surplus of 15 kg N ha⁻¹ per year in the soil content for the subsequent crop. The N balance, including atmospheric N compounds, was between 15 and 20 kg ha⁻¹ N surplus per year. NO₃⁻ levels in groundwater varied between 20 and 24 mg L⁻¹, which is below 25 mg L⁻¹ (level at which anthropogenic effects are assumed) and 50 mg L⁻¹ (limit value for groundwater in the European Union) (BMEL & BMU, 2020). The field trial area was not fertilized before sowing nor during the study period (2019–2021) to measure the effects on plant-available N in soil.

To determine the plant-available mineral N (N_{min}) content, samples were collected on the day of sowing at four depths at 0.3 m intervals (0-0.3, 0.3-0.6, 0.6-0.9, and 0.9-1.2 m) from the whole field trial area (Table S1). Samples for N_{min} determination were also collected on the first (June 27, 2019 and May 25, 2021), second (July 29, 2021), and last cutting dates (September 14, 2019, and September 14, 2021) at four depths, with four samples total along the plot width. Soil samples were taken on 2 days in 2019 and 8 days in 2021 after the last hand harvest of plant material and before plow-up. There are no N_{min} or nitrate values at later points in autumn provided in this manuscript, as this work focuses on the effect of the mixtures on soil N_{min} during the cultivation period. The soil material of each subsample and the same interval were mixed, and stones and plant material were removed. On the site, the samples were stored and transported in cool boxes at about 5°C. Afterward, they were stored at -18°C until NO₃⁻ and NH_4^+ extraction. The detailed approach for NO_3^- and NH_4^+ analysis was described by Krachunova et al. (2023). To calculate the N_{min} content in the soil, the dry matter of the soil material was determined in a drying cabinet at 105°C for 48 h. The calculations were conducted according to the following formula taken from VDLUFA (2012a) and Deutsches Institut für Normung e.V (DIN) (2005–2006):

$$N \min = \frac{NH_4 + (mg L^{-1}) \times d (dm) \times dB (g cm^{-3})}{DM \text{ soil } \times f (mL g^{-1})} + \frac{NO_3 - (mg L^{-1}) \times d (dm) \times dB (g cm^{-3})}{DM \text{ soil } \times f (mL g^{-1})}$$
(1)

where *d* is soil depth interval (0.3 m); dB is soil density in natural storage, including pore space (1.55 g cm⁻³); *f* is CaCl₂ concentration per FM (fresh matter)/soil material (2.50 mL g^{-1}).

The value of dB (soil density in natural storage, including pore space) was provided by the laboratory of the State Operating Company for Environment and Agriculture (BfUL) in Nossen, Free State of Saxony, Germany.

2.3 | Plant cultivars and sowing

The seed density of the pure stand treatments (100%) was 400 pure live seeds per m⁻² with a 0.13 m row spacing. Cultivars La Bella Campagnola (alfalfa), Libor (ribwort plantain), and Cosmolit (meadow fescue) of certified organic quality were used for both field trials in 2019 and 2021. La Bella Campagnola is a Mediterranean cultivar with high drought tolerance. The ribwort plantain cultivar Libor contains around 2% aucubin and can grow from semi-shade to no direct sunlight, according to the plant breeder Jelitto Staudensamen GmbH. Cosmolit is a drought-resistant cultivar typically used for alfalfa mixtures in Germany (Bundessortenamt, 2020).

Prior to sowing, the soil was prepared using a turning plow (Lemken, at 0.27-m depth) and a cultivator (Maschio S.p.A, at approximately 0.08-m depth). All seeds were deposited on the soil surface (Seed drill Wintersteiger Hege 80/Kombi) on April 1, 2019, and February 28, 2021. The soil surface was flattened with a roller (Prismatic roller Mediana, Güttler) after seeding. The cropping period ended on October 17, 2019, and October 21, 2021, when the fields were plowed.

2.4 | Plant sampling

Aboveground plant material was harvested on three dates: at the late bud stage of alfalfa (June 27, 2019, and May 25, 2021), at alfalfa early flowering on July 29, 2021, and at the alfalfa late flowering stage of both alfalfa and ribwort plantain (October 14, 2019, and September 14, 2021). No second cutting of plant material was possible in the 2019 field trial at the early flowering stage of alfalfa due to heavy weed infestation. Instead, the area was mowed in July 2019. Each harvest consisted of manually cutting 1.5 m² of fresh matter with hedge shears, which was then used to determine the species composition. Forage accumulation (FA) was then determined by harvesting a 7 m^{-2} strip from the center of each plot with a manually operated green fodder plot harvester (Haldrup GmbH). The material was collected immediately after cutting with rakes and the fresh matter was weighed in the field (Bosche Wägetechnik, 0.2 ± 60.0 kg). Dry matter was determined by drying until attaining constant weight at 105°C (48 h, Thermo UT 6760, +20 up to 300°C, Heraeus Holding GmbH).

The hand-cut samples were kept under cool conditions overnight (5°C) until sample processing took place the next day. Each sample was separated to quantify the mixture species fractions and weed biomass. The weeds present at the study site were assessed with the visual counting system for weed density called Göttinger Zahl- und Schätzrahmen as described by LfL (2024) on the same day of hand harvest. The fresh weights of the whole sample and of the individual forage mixture components were then determined (Denver Instrument SI-6002, accuracy ± 0.01 g).

2.5 | Statistical analysis

All statistical analyses were implemented with SAS statistical software (version 9.4 for Windows, SAS Institute Inc.). Preliminary examination of the raw datasets revealed that the FA varied greatly between the 2 years of study. Consequently, differences in the N_{min} distribution during preliminary analyses were also discovered. For this reason, the authors carried out two separate analyses of each year to represent all the effects of the mixture treatments.

All data were analyzed using PROC GLM (general linear model) for both one (treatments T mixtures and pure stands) and multi-factor (variables years 2019 and 2021 (Y), T and Y \times T) analysis of variance. PROC GLM with the least square (LS) method was chosen for the block design of the field trial as recommended by Munzert (2015). The conservative approach of Tukey's honestly significant difference (HSD, p-level set at the level as the largest mean difference) post hoc test was used for specific pairwise comparisons (Tukey, 1949). Normal distribution of all datasets was tested with the Shapiro-Wilk test (Shapiro & Wilk, 1965), whereby nonnormally distributed data were transformed according to the guidelines of Köhler et al. (2012) and Munzert (2015), the functions 1:x, $\sqrt{(x)}$, x^2 , $\sqrt{(1:x)}$, 1: x^2 , and log(x). The statistics of the data as well as the presentation were carried out with the arithmetic mean and standard deviation of the mean. The significance level for all presented results was $p \le 0.05$.

The metric unit tonnes (t) was used for the FA results and was calculated generally as follows:

t ha =
$$\frac{\text{DM yield sample (g)}}{1.5\text{m}^2 \text{ harvested plot area}} \times 100$$
 (2)

where g represents metric gram and m² represents metric square meter.

The total FA corrected for DM (dry matter) (t ha^{-1}) was used for the calculation of the partial land equivalent ratio (pLER) and total land equivalent ratio (LER). The LER was determined following the equation given by Willey and Osiru (1972) and was calculated as follows:

Partial LER

```
= \frac{\text{Forage accumulation of mixture component 1}}{\text{Forage accumulation of pure stand crop species 1}}
```

(3)

Total LER

 $= \frac{\text{Forage accumulation of mixture component species 1}}{\text{Forage accumulation of pure stand crop species 1}}$ $+ \frac{\text{Forage accumulation of mixture componen species 2}}{\text{Forage accumulation of pure stand crop species 2}}$ (4)

Significant effects of the mixture interaction between the replacement series were identified. Graphical illustrations of the results were generated with SigmaPlot (Version 12.5, Systate Software Inc.).

3 | RESULTS

3.1 | Forage accumulation and competition development

All treatments showed very low dry matter yields at the first cut in both trial years, with alfalfa showing the lowest values (p = 0.01, Table 2). The FA from treatments with forage mixtures was greater than that of alfalfa in both trial years (Table 2). Ribwort plantain with alfalfa in 50/50% mixtures led to greater FA in the first cut in 2019 compared to alfalfa in either year with all other treatments being intermediate to but not different from these two treatments. In the second cut in 2021, ribwort plantain in all alfalfa mixtures produced the greatest FA numerically, although the increase was not different from ribwort plantain, 33/67% alfalfa with meadow fescue, or 50/50% alfalfa with meadow fescue. Alfalfa had the lowest FA at the third cut in 2019 at 0.28 t ha^{-1} , which was lower than all alfalfa and ribwort plantain mixtures and ribwort plantain and meadow fescue in pure stands (p < 0.0001). Meadow fescue showed greater FA than in the mixtures with alfalfa. No statistical differences among treatments were determined at the third cut of 2021 (p = 0.59). Total FA showed similar trends in 2019 and 2021, in which ribwort plantain and ribwort plantain/alfalfa mixtures generally produced greater FA compared to alfalfa in pure stands and meadow fescue mixtures.

The most dominant weed was common poppy (*Papaver rhoeas* L.), with over 85% occurrence. The other 15% of weed occurrences consisted of purple deadnettle (*Lamium purpureum* L.), cornflower (*Centaurea cyanus* L.), common dandelion (*Taraxacum officinale* L.), common chickweed (*Stellaria media* L.), common amaranth (*Amaranthus retroflexus* L.), and single plants of red clover (*Trifolium pratense* L.), jointed charlock (*Raphanus raphanistrum* L.), and common bugloss (*Anchusa officinalis* L.) spread throughout the whole trial field.

Ribwort plantain significantly suppressed alfalfa in mixture treatments 50/50% and 67/33% at the first cut in both **TABLE 2** Forage accumulation (FA) (t ha⁻¹) of pure stands and mixture treatments of plantain, alfalfa, and meadow fescue in field trials in 2019 and 2021. The cuts were conducted as follows: 1. Cut at the late bud stage of alfalfa (June 27, 2019, and May 25, 2021), 2. cut at early alfalfa flowering (July 29, 2021; data not available for 2019), and 3. cut at the late flowering stage of alfalfa and ribwort plantain (October 14, 2019, and September 14, 2021). One-factor analysis for 2019 and 2021, and multiple comparison for 2019 × 2021 with Tukey's honestly significant difference (HSD). *p*-values indicate significance within one column of the same cut.

Year	Treatment	Seed rate (%)	1. Cut FA (t ha ⁻¹)	2. Cut FA (t ha ⁻¹) ^a	3. Cut FA (t ha ⁻¹)	Total FA (t ha ⁻¹)
2019	R. plantain	100	0.06ab	х	1.61a	1.67a
	Alfalfa/R. plantain	33/67	0.05ab	Х	1.55a	1.61a
		50/50	0.09a	Х	1.69a	1.78a
		67/33	0.05ab	Х	1.12ab	1.17ab
	Alfalfa	100	0.01b	х	0.28c	0.29c
	Alfalfa/meadow fescue	33/67	0.03ab	Х	0.71bc	0.75bc
		50/50	0.03ab	Х	0.71bc	0.74bc
		67/33	0.03ab	Х	0.68bc	0.71bc
	Meadow fescue	100	0.04ab	Х	1.19ab	1.23ab
	Tukey-HSD <i>p</i> -values		0.01	Х	<0.0001***	<0.0001***
2021	R. plantain	100	0.15ab	4.51abc	1.43	6.09abc
	Alfalfa/R. plantain	33/67	0.18ab	6.62a	1.52	8.32a
		50/50	0.27a	6.19a	1.79	8.25a
		67/33	0.16ab	5.14ab	1.37	6.67ab
	Alfalfa	100	0.01b	2.16c	1.61	3.78bc
	Alfalfa/meadow fescue	33/67	0.04b	2.13abc	1.56	3.73c
		50/50	0.10ab	2.44abc	1.48	4.02bc
		67/33	0.11ab	1.68c	1.89	3.68c
	Meadow fescue	100	0.11ab	2.05c	1.94	4.11bc
	Tukey-HSD p-values		0.01**	<0.0001***	0.59 ^{ns}	<0.0001***
2019 × 2021	Multiple comparison <i>p</i> -values	Y	<0.0001***	Х	<0.0001***	<0.0001***
		Т	0.001***	х	0.001***	<0.0001***
		$\mathbf{Y} \times \mathbf{T}$	0.15 ^{ns}	х	0.001	0.001

Note: Data for multiple comparisons are weighted means of four replications per year (Y) and treatment (T). Letters following values indicate significant differences within the same column according to a multiple comparison test. Seed rate (%) column—numbers 67/33, 50/50, and 33/67 indicate mixture proportions in percent.

Abbreviations: ns, not significant; R. plantain, ribwort plantain; T, treatment; Y, year; Y × T, interactive effect between years and treatments.

^aX's indicate daata available only for 2021; one-factor analysis with Tukey-HSD.

, *Significant at $p \le 0.01$ and $p \le 0.001$, respectively.

field trials (p < 0.0001, Table 3). Alfalfa did not establish very well at the first cut in 2019 or 2021, but it generally showed higher shares in all treatments with meadow fescue. Alfalfa showed the significantly lowest share (18.9%) compared to ribwort plantain (81.1%) in the 67/33% treatment (p < 0.0001, Table 3). Alfalfa had higher shares in ribwort plantain and meadow fescue mixture treatments during 2019 ($p_T < 0.0001$), under drier conditions, compared to 2021, which led to significant interactions between year and treatment in the multiple comparison ($p_{Y \times T} = 0.001$, Table 3).

In the second cut in 2021, the highest share of alfalfa amounted to 27.7% in the 67/33% meadow fescue treatment (p < 0.0001). Significantly lower shares of alfalfa were found in the 33/67% and 50/50% ribwort plantain mixtures. The reduction in alfalfa share in the 33/67% ribwort plantain mix-

ture was drastic, from 48.9% during the first cut to 3.1% in the second cut (Table 3).

The share of alfalfa decreased at the third cut in 2019 and 2021. The highest share of alfalfa was in the 67/33% mixture with meadow fescue in 2019 (11.4%). Alfalfa at the third cut in 2019 and 2021 was generally far below shares of ribwort plantain and meadow fescue in all treatments (p < 0.0001, Table 3). The highest share of ribwort plantain was 89.5% in 2019 and 95.1% in 2021 (Table 3).

3.2 | pLER and LER

The highest pLERs for alfalfa and ribwort plantain in 2019 and 2021 were in the 50/50% mixture, but significant only in 2019 (p = 0.002, Table 4). As a result, the highest total LERs in the same treatment were significant in 2019 (p = 0.01, Table 4).

ompetition development in the share of alfalfa, ribwort plantain, and meadow fescue (%) in the mixture treatments in field trials in 2019 and 2021. The cuts were conducted as follows:	ud stage of alfalfa (June 27, 2019, and May 25, 2021), 2. cut at early alfalfa flowering (July 29, 2021; data not available for 2019), and 3. cut at the late flowering stage of alfalfa and	October 14, 2019, and September 14, 2021). One-factor analysis for 2019 and 2021, and multiple comparison for 2019 × 2021 with Tukey's honestly significant difference (HSD).	significance within two columns of the same cut.
ABLE 3 Competition develop	. Cut at the late bud stage of alfalfa (bwort plantain (October 14, 2019, a	-values indicate significance within

				1. Cut		2. Cut		3. Cut		Mean
		Seed rate	1. Cut alfalfa	R. plantain/ M. fescue	2. Cut alfalfa	R. plantain/ M. fescue	3. Cut alfalfa	R. plantain/ M. fescue	Mean Alfalfa	R. plantain/ M. fescue
Year	Treatment	(%)	share (%)	share (%)	share (%) ^a	share (%) ^a	share (%)	share (%)	share (%)	share (%)
2019	Alfalfa/R. plantain	33/67	28.33cd	71.67ab	х	×	1.34d	98.66a	14.84b	85.16a
		50/50	28.32cd	71.68ab	х	х	1.17d	98.83a	14.74b	85.26a
		67/33	18.88d	81.12a	x	X	2.08d	97.92a	10.48b	89.52a
	Alfalfa/meadow fescue	33/67	41.28bcd	58.72abc	×	×	2.22d	97.78a	21.75b	78.25a
		50/50	43.48abcd	56.52abcd	×	×	9.49c	90.51b	26.49b	73.51a
		67/33	47.77abcd	52.23abcd	х	х	11.37c	88.63b	29.57b	70.43a
	Tukey-HSD <i>p</i> -values		$<0.0001^{***}$		×		<0.0001***		<0.0001***	
2021	Alfalfa/R. plantain	33/67	48.90ab	51.10ab	3.10d	96.90a	2.99b	97.01a	18.33b	81.67a
		50/50	6.59b	93.41a	4.29d	95.71a	3.92b	96.08a	4.93b	95.07a
		67/33	12.50b	87.50a	15.58cd	84.42ab	10.58b	89.42a	12.89b	87.11a
	Alfalfa/meadow fescue	33/67	39.53ab	60.47ab	12.55cd	87.45ab	4.72b	95.28a	18.93b	81.07a
		50/50	20.58b	79.42a	15.88cd	84.12ab	7.47b	92.53a	14.65b	85.35a
		67/33	3.96b	96.04a	27.73c	72.27b	10.25b	89.75a	13.98b	86.02a
	Tukey-HSD p-values		<0.0001***		<0.0001***		<0.0001***		<0.0001***	
2019 × 2021	Multiple comparison <i>p</i> -values	Y	1.0 ^{ns}		x		1.0 ^{ns}		1.0 ^{ns}	
		Т	$<0.0001^{***}$		<0.0001***		<0.0001***		<0.0001***	
		$\mathbf{Y}\times\mathbf{T}$	0.001		×		0.01		0.01	

Note: Data in the multiple comparisons are weighted means of four replications per year (Y) and treatment (T). Letters following values indicate significant differences within the same column according to a multiple comparison test. Seed rate (%) column-Numbers 67/33, 50/50, and 33/67 indicate mixture proportions in percent.

Abbreviations: ns, not significant; M. fescue, meadow fescue; R. plantain, ribwort plantain; T, treatment; Y, year; Y × T, interactive effect between years and treatments.

^aX's indicate data available only for 2021, one-factor analysis with Tukey-HSD.

***Significant at $p \leq 0.01$ and 0.001.

8 of 20

				1. Cut			2. Cut pLER			3. Cut			Total	
		Seed	1. Cut	pLER	1. Cut	2. Cut	R.	2. Cut	3. Cut	pLER	3. Cut	Total	pLER	
Year	Treatment	rate (%)	pLER Alfalfa	R. plantain/ M. fescue	Total LER	pLER Alfalfa ^a	plantain/M. fescue	Total LER ^a	pLER Alfalfa	R. plantain/ M. fescue	Total LER	pLER Alfalfa	R. plantain/ M. fescue	Total LER
2019	Alfalfa/R. Plantain	33/67	0.86abc	0.74abc	1.60b	X	X	×	0.06b	1.01a	1.07	0.57b	0.88ab	1.45ab
		50/50	1.72a	1.15abc	2.87a	×	Х	×	0.07b	1.07a	1.14	1.59a	1.11ab	2.70a
		67/33	1.51ab	0.65bc	2.16ab	Х	Х	X	0.10b	0.74ab	0.84	0.57b	0.69b	1.26b
	Alfalfa/meadow fescue	33/67	1.02abc	0.49bc	1.51b	Х	Х	X	0.11b	0.63ab	0.74	0.78ab	0.56b	1.34ab
		50/50	1.12abc	0.45bc	1.57b	Х	Х	Х	0.27ab	0.57ab	0.84	0.70b	0.51b	1.21b
		67/33	0.99abc	0.36c	1.35b	X	X	×	0.58ab	0.55ab	1.13	0.79ab	0.45b	1.24b
	Tukey-HSD p-values		0.002^{**}		0.01^{**}	X		×	0.001^{***}		0.52	0.003**		0.01^{**}
2021	Alfalfa/R. plantain	33/67	1.41	0.65	2.06	0.09e	1.42a	1.52	0.04d	1.07a	1.11	0.68bc	1.05ab	1.73
		50/50	1.52	1.63	3.15	0.12e	1.32ab	1.44	0.04d	1.23a	1.27	0.56bc	1.39a	1.95
		67/33	0.82	0.92	1.74	0.36de	0.97abc	1.33	0.08c	0.87abc	0.95	0.42bc	0.92abc	1.34
	Alfalfa/meadow fescue	33/67	1.50	0.50	2.00	0.17e	0.64cde	0.81	0.04d	0.83abcd	0.87	0.45bc	0.66bc	1.11
		50/50	1.77	0.43	2.20	0.31de	0.86bcd	1.17	0.06d	0.73abcd	0.79	0.72bc	0.68bc	1.40
		67/33	0.45	0.92	1.37	0.53cde	0.48cde	1.01	0.13bcd	0.91ab	1.04	0.37c	0.77abc	1.14
	Tukey-HSD p-values		0.2 ^{ns}		0.48^{ns}	<0.0001***		0.15 ^{ns}	<0.0001**	*	0.59 ^{ns}	0.001***		0.17 ^{ns}
2019 × 2021	<i>p</i> -values	Y	0.32 ^{ns}		0.18 ^{ns}	×		×	0.63 ^{ns}		0.57 ^{ns}	0.5 ^{ns}		0.34 ^{ns}
		T	0.001***		0.1 ^{ns}	<0.0001***		0.15 ^{ns}	<0.0001**	*	0.23 ^{ns}	0.001***		0.01
		$\mathbf{Y} \times \mathbf{T}$	0.49 ^{ns}		0.98 ^{ns}	×		Х	0.4 ^{ns}		0.96 ^{ns}	0.01		0.89 ^{ns}

The cuts were conducted as follows: 1. Cut at the late bud stage of alfalfa (June 27, 2019, and May 25, 2021), 2. cut at early alfalfa flowering (July 29, 2021; data not available for 2019), and 3. cut at the late flowering stage of alfalfa and ribwort plantain (October 14, 2019, and September 14, 2021). One-factor analysis for 2019 and 2021, and multiple comparison for 2019 × 2021 with Tukey's honestly Partial land equivalent ratio (pLER) and total land equivalent ratio (total LER) of ribwort plantain, alfalfa, and meadow fescue in the mixture treatments in field trials in 2019 and 2021. TABLE 4

test.

^aX's indicate data available only for 2021, one-factor analysis with Tukey-HSD. Seed rate (%) column—Numbers 67/33, 50/50, and 33/67 indicate mixture proportions in percent. Abbreviations: ns, not significant; M. fescue, meadow fescue; R. plantain; ribwort plantain; T, treatment; Y, year; Y × T, interactive effect between years and treatments.

***,**Significant at $p \leq 0.01$ and 0.001, respectively.

Agronomy Journal



FIGURE 2 Plant-available mineral nitrogen (N_{min}) content (kg N ha⁻¹) at a 0- to 0.3-m soil depth in field trials in 2019 and 2021: (A) on June 27, 2019, p = 0.06; and (B) on May 25, 2021, p = 0.66. One-factor analysis with Tukey's honestly significant difference (HSD). Means \pm standard deviation. Numbers 67/33, 50/50, and 33/67 indicate mixture proportions in percent, and 100 indicates a pure stand. AA, alfalfa pure stand; MF, meadow fescue; MF/AA, meadow fescue/alfalfa mixtures; RP, ribwort plantain pure stand; RP/AA, ribwort plantain/alfalfa mixtures.

There were no significances in 2021 (pLER p = 0.2, total LER p = 0.48, Table 4). For both pLER and total LER, there were no significant interactions between year and treatment (pLER $p_{\text{Y} \times \text{T}} = 0.49$, total LER $p_{\text{Y} \times \text{T}} = 0.98$, Table 4).

For the second cut in 2021, a significant decline in the pLER of alfalfa was observed in all treatments (p < 0.0001, Table 4). pLER of ribwort plantain was 33/67% and was significantly higher than all other pLERs (p < 0.0001, Table 4). The total LER in the second cut was numerically higher among alfalfa/ribwort plantain than alfalfa/meadow fescue treatments in 2021, but not significant (p = 0.15, Table 4).

In the third cut in 2019 and 2021, the highest pLER was achieved by ribwort plantain in the 50/50% treatment with alfalfa (p = 0.001 in 2019 and p < 0.0001 in 2021, Table 4). The pLER values of alfalfa were significantly lower than the pLER of ribwort plantain in 33/67% and 50/50% mixtures in 2019 and 2021 (p = 0.001 in 2019, p < 0.001 in 2021, Table 4). The total LER of the treatments showed no significances in the one-factor analysis or significant interactions in the multiple comparison (p = 0.52 in 2019, p = 0.52 in 2021, $p_{Y \times T} = 0.96$, Table 4).

Total LER over three cuts was significantly highest for 50/50% alfalfa/ribwort plantain treatments in 2019 (p = 0.01, Table 4). There were no significant differences for the total LER in 2021 (p = 0.17, Table 4). Overall, a higher total LER without significant interactions of year × treatment was achieved under the ribwort plantain treatments across all cuts and years ($p_T = 0.01$, $p_{Y \times T} = 0.89$, Table 4).

3.3 | N_{min} content

At 0- to 0.3-m soil depth, no clear trends in N_{min} development were observed for the first cut in either trial year (Figure 2A,B). Higher values were recorded in 2019 for ribwort plantain (10.6 kg N ha⁻¹) and alfalfa (10.8 kg N ha⁻¹) in pure stands, but not significantly (p = 0.06, Figure 2A). In 2021, the 67/50% and 50/50% ribwort plantain/alfalfa mix-

tures, as well as alfalfa in pure stands, showed the greatest standard deviations and no significance (p = 0.66, Figure 2B).

On the autumn date in 2019, N_{min} of 100% alfalfa showed significant differences at 0- to 0.3-m depth from 100% ribwort plantain and 67/33% ribwort plantain/alfalfa treatments (p = 0.03, Figure 3A). The significantly highest N_{min} at 0.3- to 0.6-m soil depth was determined under 100% alfalfa in 2019 and there was a trend for higher N_{min} in meadow fescue/alfalfa treatments (p = 0.01, Figure 3B). Significant differences among the N_{min} were also determined from 0.6 to 0.9 m in 2019, with 100% meadow fescue having the highest mean (p = 0.01, Figure 3C). From 0.9 to 1.2 m, the trend for higher N_{min} content under 100% alfalfa and alfalfa/meadow fescue treatments remained visible, but without a significance (p = 0.16, Figure 3D).

In autumn 2021, the N_{min} values under the presence of ribwort plantain, to the left of pure-cultured alfalfa in Figure 4A showed almost equal Nmin values with a low standard deviation, while the values for alfalfa in pure stands and meadow fescue mixtures showed a large dispersion. There were no significances, as shown in Figure 4A (p = 0.11). The highest N_{min} value in 2021 was also found at 0.3-0.6 m, as in 2019, under 100% alfalfa (p = 0.03, Figure 4B). The 50/50% and 33/67% ribwort plantain/alfalfa mixtures showed significantly lower N_{min} contents. From 0.6- to 0.9-m depth, the N_{min} values showed a greater scatter, with 33/67% meadow fescue/alfalfa treatment having the highest value. The lowest value was found under 100% ribwort plantain, but not significant (p = 0.07, Figure 4C). From 0.9 to 1.2 m, the individual values of the treatments were scattered and not significant, but higher N_{min} values were found with meadow fescue/alfalfa treatments (p = 0.4, Figure 4D).

3.4 | NO₃-N and NH₄-N share

Analysis of the mean values showed that the significantly lowest NO_3 -N share (65.9%) was present in the pure stand



FIGURE 3 Plant-available mineral nitrogen (N_{min}) content (kg N ha⁻¹) at different soil depths in the 2019 field trial on October 16, 2019: (A) 0–0.3 m, p = 0.03; (B) 0.3–0.6 m, p = 0.01; (C) 0.6–0.9 m, p = 0.01; (D) 0.9–1.2 m, p = 0.16. One-factor analysis with Tukey's honestly significant difference (HSD). Means ± standard deviation. Lowercase letters a and b indicate significance within the same sub-figure. Numbers 67/33, 50/50, and 33/67 indicate mixture proportions in percent, and 100 indicates a pure stand. AA, alfalfa pure stand; MF, meadow fescue; MF/AA, meadow fescue/alfalfa mixtures; RP, ribwort plantain pure stand; RP/AA, ribwort plantain/alfalfa mixtures.

of ribwort plantain (p = 0.02, Figure 5A). The 33/67% ribwort plantain/alfalfa treatment showed the significantly highest proportion of NO₃-N (85.6%) at 0–0.3 m. Conversely, 100% ribwort plantain had the significantly highest NH₄-N share in pure stands and the significantly lowest share in the 33/67% treatment with alfalfa. No significant differences were observed from 0 to 0.3 m at the first cutting date in May 2021 (p = 0.76, Figure 5B). In contrast to the cut in 2019, 100% alfalfa showed the lowest NO₃-N content of all treatments. Ribwort plantain and meadow fescue in pure stands showed similar values (Figure 5B).

There were no significant differences in NO₃-N and NH₄-N from 0- to 0.3-m depth during the second cut in mid-October 2019 (p = 0.14, Figure 6A). However, the treatments with 33/67% and 50/50% meadow fescue/alfalfa had the lowest proportion of NO₃-N and consequently the highest proportion of NH₄-N. At 0.3- to 0.6-m soil depth, the 67/33% ribwort plantain/alfalfa treatment had the significantly lowest NO₃-N ratio and highest NH₄-N ratio compared to 100% alfalfa and 33/67% meadow fescue/alfalfa treatment (p = 0.01, Figure 6B). From 0.6- to 0.9-m depth, the NO₃-N proportion was higher in 100% alfalfa and meadow fescue and all meadow fescue/alfalfa treatment than in 100% ribwort plantain, but not significant (p = 0.1, Figure 6C). The highest NO₃-N shares occurred in 33/67% and 50/50% alfalfa/meadow fescue mixtures. The NH₄-N percentage in pure ribwort plantain was slightly lower compared to all ribwort plantain/alfalfa treatments (Figure 6C). At 0.9- to 1.2-m depth, the NH₄-N ratio decreased in all treatments, except for the 67/33% ribwort plantain/alfalfa mixture. There were outliers in the datasets for this and 67/33% meadow fescue/alfalfa treatments, which is why it was difficult to interpret the results (p = 0.16, Figure 6D).

At the second cutting date in mid-September 2021, the NO₃-N share at 0-0.3 m was significantly higher with 100% alfalfa and 33/67% meadow fescue/alfalfa in contrast to the 33/67% ribwort plantain/alfalfa treatment (p = 0.04, Figure 7A). From 0.3 to 0.6 m, the significantly lowest NO₃-N share was observed with pure ribwort plantain (p = 0.003, Figure 7B). Pure cultured alfalfa and the 50/50% meadow fescue/alfalfa treatments had similar values from 0.3 to 0.6 m (Figure 7B). From 0.6 to 0.9 m, the difference between the NO₃-N /NH₄-N ratio with pure alfalfa and the 33/67% meadow fescue/alfalfa mixture clearly increased, but without a significance (p = 0.09, Figure 7C). With 100% alfalfa, the NO₃-N share was 37.4% higher than in the 33/67%ribwort plantain/alfalfa mixture compared with 0.3-0.6 m. Furthermore, Figure 7C reveals that all treatments containing ribwort plantain (left side of 100% alfalfa in Figure 7C) had higher NH₄-N shares than all treatments with meadow



FIGURE 4 Plant-available mineral nitrogen (N_{min}) content (kg N ha⁻¹) at different soil depths in the 2021 field trial on September 14, 2021: (A) 0–0.3 m, p = 0.11; (B) 0.3–0.6 m, p = 0.03; (C) 0.6–0.9 m, p = 0.07; and (D) 0.9–1.2 m, p = 0.4. One-factor analysis with Tukey's honestly significant difference (HSD). Means ± standard deviation. Lowercase letters a and b indicate significance within the same sub-figure. Numbers 67/33, 50/50, and 33/67 indicate mixture proportions in percent, and 100 indicates a pure stand. AA, alfalfa pure stand; MF, meadow fescue; MF/AA, meadow fescue/alfalfa mixtures; RP, ribwort plantain pure stand; RP/AA, ribwort plantain/alfalfa mixtures.



FIGURE 5 NO₃-N to NH₄-N (%) share in total N_{min} at a 0- to 0.3-m soil depth in field trials in 2019 and 2021: (A) 0–0.3 m on June 27, 2019, $p_{NO_3-N} = 0.02$, $p_{NH_4-N} = 0.02$; and (B) 0–0.3 m on May 25, 2021, $p_{NO_3-N} = 0.76$, $p_{NH_4-N} = 0.76$. One-factor analysis with Tukey's honestly significant difference (HSD). Means ± standard deviation. Lowercase letters a and b indicate the significance of NO₃-N (%) within the same sub-figure. Uppercase letters A and B indicate the significance of NH₄-N (%) within the same sub-figure. Numbers 67/33, 50/50, and 33/67 indicate mixture proportions in percent, and 100 indicates a pure stand. AA, alfalfa pure stand; MF, meadow fescue; MF/AA, meadow fescue/alfalfa mixtures; RP, ribwort plantain pure stand; RP/AA, ribwort plantain/alfalfa mixtures.

fescue (right of 100% alfalfa in Figure 7C). From 0.9 to 1.2 m, the NO₃-N shares increased in the 33/67% and 50/50% meadow fescue/alfalfa mixtures and were significantly higher than 67/33% and 33/67% ribwort plantain/alfalfa mixtures (p = 0.001, Figure 7D).

4 | DISCUSSION

The authors showed for the first time that a ribwort plantain/alfalfa mixture increased the share of NH_4 -N in annual alfalfa mixtures in an autumn plowing. Although the effect



FIGURE 6 NO₃-N to NH₄-N (%) share in total N_{min} at different soil depths in the 2019 field trial on October 16, 2019: (A) 0–0.3 m, $p_{NO_3-N} = 0.14$, $p_{NH_4-N} = 0.14$; (B) 0.3–0.6 m, $p_{NO_3-N} = 0.01$, $p_{NH_4-N} = 0.01$; (C) 0.6–0.9 m, $p_{NO_3-N} = 0.1$, $p_{NH_4-N} = 0.1$; (D) 0.9–1.2 m, $p_{NO_3-N} = 0.16$, $p_{NH_4-N} = 0.16$. One-factor analysis with Tukey's honestly significant difference (HSD). Means \pm standard deviation. Lowercase letters a and b indicate the significance of NO₃-N (%) within the same sub-figure. Uppercase letters A and B indicate the significance of NH₄-N (%) within the same sub-figure. Numbers 67/33, 50/50, and 33/67 indicate mixture proportions in percent, and 100 indicates a pure stand. AA, alfalfa pure stand; MF, meadow fescue; MF/AA, meadow fescue/alfalfa mixtures; RP, ribwort plantain pure stand; RP/AA, ribwort plantain/alfalfa mixtures.

of ribwort plantain on the soil was observed in both trial years, the yields and shares of mixture plants from all mixture species and the weather conditions in both trial years varied greatly. The outlined results show the tested mixture and pure stands of alfalfa, ribwort plantain, and meadow fescue under precipitation amounts that were below the long-term mean in 2019, as well as under precipitation amounts that deviated from the long-term mean in 2021.

4.1 | Ribwort plantain enhances forage accumulation

Ribwort plantain developed vigorously and thus contributed negatively to alfalfa growth (Tables 2 and 3). The ribwort plantain is a very competitive plant species that has moderate demands for site conditions (Ellenberg & Leuschner, 2010). A pot experiment with ribwort plantain showed that even under competition stress and lack of nutrient supply, it did not initiate reproduction and did not show higher costs of chemical defense; in contrast, the aboveground dry matter losses were less than under favorable environmental condi-

tions (Marak et al., 2003). Ribwort plantain developed faster than alfalfa and meadow fescue in the cooler spring of 2021, while in the warmer spring of 2019, the development of both ribwort plantain and meadow fescue was roughly the same. According to Marak et al. (2003), this can be explained by the fact that ribwort plantain grows well under cool temperatures. Alfalfa is susceptible to light competition from weed plants during its development (Veronesi et al., 2010). Alfalfa is a drought-resistant plant that, similar to ribwort plantain (Pol et al., 2021), has a deep root system. However, the roots of young alfalfa plants are not yet sufficiently developed to withstand drought stress as they cannot reach deep soil layers and can be damaged (Prince et al., 2022; Sheaffer et al., 1988). Under dry conditions, N-uptake is inhibited in grasses due to drought stress, as the growth rate decreases (Burity et al., 1989). Similar cultivation methods for alfalfa in the literature have thus far achieved a higher FA in the first year of cultivation compared to the present trial with 2 t ha^{-1} in three cuts. Over three cuts, Rakotovololona et al. (2019) reached an alfalfa FA of 8.9 t ha^{-1} in the first growing year. Masoni et al. (2015) found an alfalfa yield of 15 t ha^{-1} over five cuts in the first growing year. Koch (1997) reported an



FIGURE 7 NO₃-N to NH₄-N (%) share in total N_{min} at different soil depths in the 2021 field trial on September 14, 2021: (A) 0–0.3 m, $p_{NO_3-N} = 0.04$, $p_{NH_4-N} = 0.04$; (B) 0.3–0.6 m, $p_{NO_3-N} = 0.003$, $p_{NH_4-N} = 0.003$; (C) 0.6–0.9 m, $p_{NO_3-N} = 0.09$, $p_{NH_4-N} = 0.09$; and (D) 0.9–1.2 m, $p_{NO_3-N} = 0.001$, $p_{NH_4-N} = 0.001$. One-factor analysis with Tukey's honestly significant difference (HSD). Means ± standard deviation. Lowercase letters a and b indicate significance of NO₃-N (%) within the same sub-figure. Uppercase letters A and B indicate significance of NH₄-N (%) within the same sub-figure. Numbers 67/33, 50/50, and 33/67 indicate mixture proportions in percent, and 100 indicates a pure stand. AA, alfalfa pure stand; MF, meadow fescue; MF/AA, meadow fescue/alfalfa mixtures; RP, ribwort plantain pure stand; RP/AA, ribwort plantain/alfalfa mixtures.

8.8–11.3 t ha⁻¹ dry matter yield on clayey silt. A practical test of the same alfalfa variety under similar soil and climatic conditions in Germany in 2021 showed FA of 2.5 t ha⁻¹ over two cuts in annual cultivation, with very low values for the first cut, similar to the present trial (Kling & Bruckner, 2024).

The FA of the 67/33% and 50/50% ribwort plantain/alfalfa treatments were significantly higher at the second (2021) and the third cutting dates (2019, 2021) compared to the reference alfalfa/meadow fescue mixtures (Table 2). Skinner and Gustine (2002) found that the ribwort plantain yield was greater in September than in July, increasing by 62% in the control and 29% in a watered treatment. The imposition of summer drought on ribwort plantain increased winter survival from 3% in the wet treatment to 41% in the dry treatment (Skinner & Gustine, 2002). Rumball et al. (1997) also reported higher yields in autumn. However, most of the yield (>90%) was accounted for by the dry matter yield of ribwort plantain. The FA of ribwort plantain doubled over three cuts, both under dry conditions (2019) and higher precipitation (2021). Dembek et al. (2015) observed the effects of ribwort plantain on the yield and quality of sward in grasslands between

2006 and 2011 and found that ribwort plantain produced the highest grassland yields compared to the other species studied under dry weather conditions. In this study, ribwort plantain produced the highest FA in pure stand and with proportions between 84% and 97% in the mixture treatments with alfalfa.

The aucubin in ribwort plantain forage has an antimicrobial effect on ruminants. Ribwort plantain cuttings are well degradable in the rumen of ruminants and promote milk production (Mangwe et al., 2020). Ribwort plantain is suitable as an additive for sheep fodder because it significantly lowers the number of certain parasites in the feces (Reza et al., 2021). Furthermore, the urine of ribwort plantain-fed cattle reduces NO₃-N levels in soil and nitrous oxide emissions on pastures (Judson et al., 2019; Peterson et al., 2022; Simon et al., 2019).

The aucubin content varies in ribwort plantain depending on the time of harvest. Before flowering, the aucubin content is very low in every plant organ and reaches its maximum in autumn (European Medicines Agency, 2011). The new leaves of ribwort plantain contain more N and less aucubin (Bowers & Stamp, 1992). The aucubin content of leaves increases during the growing period of ribwort plantain but remains stable in the main growing period in the summer (Tamura & Nishibe, 2002). Darrow and Deane Bowers (1997) found a 0.5% aucubin content in leaf dry matter harvested in July and 4%-5% after harvests in September and October. Dietz et al. (2013) recorded 0.5%-1% aucubin in July, 5% in August, and 3%-4% in September.

Results from this work can inform organic farming practices, not only with regard to securing FA under dry conditions but also as a recommendation for preventive groundwater protection. The results obtained here also indicate a recommendation for ribwort plantain sowing density in mixtures with alfalfa due to the competitive growth pattern of ribwort plantain. A 25% or lower sowing density of ribwort plantain is recommended in a mixture with alfalfa. Depending on the 1000-grain mass of ribwort plantain seed, this leads to recommended seed rates of $0.75-1.0 \text{ kg ha}^{-1}$ in the mixture. Based on our observations, if the sowing rate of ribwort plantain is reduced to about 25% and more than one forage legume species is added to the mixture, such as white clover (Trifolium repens L.) or red clover (Trifolium pratense L.), it can be assumed that the FA of the legumes in the mixtures will increase. Mixtures of more than two forage legume species showed higher yield performance (Kirwan et al., 2007), as the legume dry matter proportion in the total mixture yield should be between 30% and 40% for high quality forage for ruminants (Broderick, 1995; Nußbaum, 2007). A further recommendation is to sow ribwort plantain to a depth of 0.5-1 cm (loamy soils) or 1-1.5 cm (sandy soils) due to its small seed size. Furthermore, the cutting time of the ribwort plantain must not be too late, as it tends to form seeds quickly, which could become problematic in the subsequent organic crop rotation. As both alfalfa and ribwort plantain can use water reserves in the lower soil layers due to their deep roots (Pol et al., 2021; Sheaffer et al., 1988), caution should be taken in water protection areas to ensure that groundwater formation is not impaired. Further research is needed in this area. In this regard, a comparison of the yield performance of different ribwort plantain varieties under field conditions would be particularly useful for organic farming.

4.2 | NO₃-N shifts to NH₄-N with ribwort plantain

Although the ribwort plantain/alfalfa mixtures were very heavily weeded in the first trial, the presence of alfalfa even in small quantities, affected the plant-available N in its pure strand and in the treatments with meadow fescue. The results of the NO₃-N and NH₄-N ratio in autumn showed more NO₃-N under 100% alfalfa, which agreed with the findings of Dollete et al. (2024) and Wery et al. (1986), as dry conditions do not inhibit the nitrification of N. The symbiotic N fixation of alfalfa depends on water availability in the soil (Dollete et al., 2024; Wery et al., 1986). However, the mechanisms by which alfalfa adapts to environmental stress, in particular to drought, are still not well known (Soba et al., 2019). The root nodules in alfalfa roots go through anaerobiosis during drought periods with high water stress, which decreases oxygen uptake in the soil and affects the nodules (Becana et al., 1986). Dollete et al. (2024) showed that drought conditions led to lower biomass and reduced symbiotic N fixation in alfalfa in a glasshouse experiment. By damaging roots and nodules, they may release organic N compounds, which are converted into NH₄-N and subsequently into NO₃-N through nitrification (Dollete et al., 2024). Other studies have concluded that drought periods lead to a boost in the alfalfa yield (Athar & Johnson, 1996). High levels of environmental stress can lead to subsequent nodule growth in forage legumes as an evolutionary attempt to sustain or compensate for the efficiency of the plant (Bordeleau & Prévost, 1994). Alfalfa roots usually expand their growth (width and depth) in soil during drought periods and increase their N concentration (Antolín et al., 1995; Soba et al., 2019). The first differences in the plant-available N composition in the shares of NO₃-N and NH₄-N were observed at the first cutting date at the end of June 2019. These results can be explained by the weather conditions in 2019, as the time from sowing to the first cutting date was 1 month longer compared to 2021. Differences were observed in June 2019, as the aucubin content in ribwort plantain was likely higher than in May 2021. The highest aucubin concentration is found in intermediate-aged leaves and the lowest in mature leaves of ribwort plantain (Bowers & Stamp, 1992). The experiments in this study were carried out under generally unfavorable soil and climate conditions with a high ribwort plantain sowing rate, whereby alfalfa was heavily displaced by ribwort plantain. According to Burity et al. (1989), up to 50% of the total N in grass in alfalfa mixtures could be derived from N fixation. With a higher share of alfalfa in a mixture, fixing more N, the nitrification inhibition of ribwort plantain was more pronounced in soil.

In October (Figures 6 and 7), a clearer NO_3 -N shift to NH_4 -N was observed at 0.3- to 1.2-m soil depth with ribwort plantain/alfalfa mixtures, as influenced by the presence of ribwort plantain. The relationship between the initially sown proportion of ribwort plantain seeds, the ribwort plantain growth, and the residual NO_3 -N reserves in the soil was not clearly recognizable. Ribwort plantain did not go to seed during 2019 and 2021, but lower seed proportions tended to accumulate more forage in mixtures. Ribwort plantain had generally higher FA with low NH_4 -N shares in 50/50% alfalfa treatment in autumn in 2019 and 2021 (Table 2; Figures 6 and 7).

In our field trials, we used a ribwort plantain cultivar that generally has a high aucubin content. However, there are other ribwort plantain cultivars that may not have the same effects under different soil types and drier climate conditions. Pol

KRACHUNOVA ET AL.

et al. (2024) showed that yield performance and aucubin content vary among commercial cultivars. Furthermore, Pol et al. (2024) found that the aucubin content varies under different light, soil, and water conditions.

5 | CONCLUSION

We showed that ribwort plantain has the overall potential to contribute to nitrate reduction and yield enhancement. Despite the significant annual differences, the same effect of an overall nitrate reduction was observed. We had a site with challenging conditions for agriculture in a water-protection area with strict rules for agricultural management. Therefore, based on our results, we conclude that ribwort plantain has a nitrate-reducing effect. We would expect the same effect, weaker or more pronounced, to occur in a new field trial with better soil conditions. However, it must also be emphasized that the influences of the different weather conditions should be more thoroughly investigated under different soil conditions.

FA was highest, with the presence of ribwort plantain, both in pure stands and mixtures. Ribwort plantain outperformed alfalfa with a share of >80% in all mixture treatments, with a tendency to increase FA by early autumn. For this reason, it is advisable for practitioners to greatly reduce the proportion of ribwort plantain seed and maintain it below 25% of its pure seed rate in mixtures with alfalfa or other forage legumes. A combination of ribwort plantain with, for example, two forage legume species would be conceivable in practice under dry conditions to increase the share of legumes in the mixture. In both field trials, the meadow fescue/alfalfa mixtures resulted in lower FA for meadow fescue compared to pure stands. However, due to the limitations of the study, it should be emphasized that the effects presented might not be the same under, for example, clay soils or even drier locations.

In autumn, the NO₃-N content at 0–0.3 and 0.3–0.6 m was higher in the treatments with 100% alfalfa and meadow fescue and their mixtures compared to ribwort plantain. The ratio gap continued to increase from 0.6 to 0.9 m, as ribwort plantain clearly affected the NO₃-N share, as all treatments with it had higher NH₄-N shares.

As a new crop rotation element, ribwort plantain can directly promote the resource efficiency of an agroecosystem, as the cultivation is based solely on the utilization of natural resources and their interactions. Furthermore, the introduction of ribwort plantain to agroecological systems can help farmers cope with the adverse effects of climate change. This study can also help farmers in water protection areas to reduce the risk of agricultural cultivation restrictions for forage legumes. Additionally, with meadow fescue as a less popular grass species, the study expanded the number of plant species on agricultural land.

AUTHOR CONTRIBUTIONS

Tsvetelina Krachunova: Conceptualization; formal analysis; investigation; methodology; project administration; visualization; writing—original draft. **Sonoko Bellingrath-Kimura**: Supervision; validation; writing—review and editing. **Knut Schmidtke**: Conceptualization; funding acquisition; methodology; project administration; supervision; validation; writing—review and editing.

ACKNOWLEDGMENTS

The authors thank Hannah Scharfstädt and Martin Scholz from University of Applied Sciences Dresden, who were involved in the field work, plant and soil sampling, and sorting in 2019. The authors would also like to thank Wassergut Canitz GmbH of public utility Leipzig for providing the test areas and Ms. Caroline Golatowski and Dr. Bernhard Wagner for their active support of the research project.

Open access funding enabled and organized by Projekt DEAL.

CONFLICT OF INTEREST STATEMENT The authors declare no conflicts of interest.

DATA AVAILABILITY STATEMENT

The original data that support the findings of this original research manuscript are available on request from the corresponding author.

ORCID

Tsvetelina Krachunova b https://orcid.org/0009-0000-6285-8756

Sonoko Bellingrath-Kimura Dhttps://orcid.org/0000-0002-6457-3583

Knut Schmidtke b https://orcid.org/0000-0002-1942-5602

REFERENCES

- Antolín, M. C., Yoller, J., & Sánchez-Díaz, M. (1995). Effects of temporary drought on nitrate-fed and nitrogen-fixing alfalfa plants. *Plant Science*, 107(2), 159–165. https://doi.org/10.1016/0168-9452(95) 04108-7
- Aponte, A., Samarappuli, D., & Berti, M. T. (2019). Alfalfa–grass mixtures in comparison to grass and alfalfa monocultures. *Agronomy Journal*, 111(2), 628–638. https://doi.org/10.2134/agronj2017. 12.0753
- Athar, M., & Johnson, D. A. (1996). Nodulation, biomass production, and nitrogen fixation in alfalfa under drought. *Journal of Plant Nutrition*, 19(1), 185–199. https://doi.org/10.1080/01904169609365116
- Barbieri, P., Pellerin, S., Seufert, V., Smith, L., Ramankutty, N., & Nesme, T. (2021). Global option space for organic agriculture is delimited by nitrogen availability. *Nature Food*, 2(5), 363–372. https://doi.org/10.1038/s43016-021-00276-y
- Becana, M., Aparicio-Tejo, P. M., & Sánchez-Díaz, M. (1986). Nitrate metabolism in alfalfa root nodules under water stress. *Journal of Experimental Botany*, 37(6), 798–806. https://doi.org/10.1093/jxb/ 37.6.798

- Benoit, M., Garnier, J., Anglade, J., & Billen, G. (2014). Nitrate leaching from organic and conventional arable crop farms in the Seine Basin (France). *Nutrient Cycling in Agroecosystems*, 100(3), 285–299. https://doi.org/10.1007/s10705-014-9650-9
- BMEL, & BMU. (2020). Nitratbericht 2020: Gemeinsamer Bericht der Bundesministerien für Umwelt, Naturschutz und nukleare Sicherheit sowie für Ernährung und Landwirtschaft. https://www.bmuv.de/fileadmin/Daten_BMU/Download_PDF/ Binnengewaesser/nitratbericht_2020_bf.pdf
- Bordeleau, L. M., & Prévost, D. (1994). Nodulation and nitrogen fixation in extreme environments. *Plant and Soil*, *161*(1), 115–125. https://doi. org/10.1007/BF02183092
- Bowers, M. D., & Stamp, N. E. (1992). Chemical variation within and between individuals of *Plantago lanceolata* (Plantaginaceae). *Journal of Chemical Ecology*, 18(7), 985–995. https://doi.org/10.1007/ BF00980057
- Broderick, G. A. (1995). Desirable characteristics of forage legumes for improving protein utilization in ruminants. *Journal of Animal Science*, 73(9), 2760–2773. https://doi.org/10.2527/1995.7392760x
- Bundesministerium der Justiz und für Verbraucherschutz. (2009). Gesetz zur Ordnung des Wasserhaushalts (Wasserhaushaltsgesetz—WHG). https://www.gesetze-im-internet.de/whg_2009/BJNR258510009. html
- Bundessortenamt. (2020). Beschreibende Sortenliste Futtergräser: Ersparsette, Klee, Luzerne. https://www.bundessortenamt.de/bsa/ media/Files/BSL/bsl_futtergraeser_2020.pdf
- Burity, H. A., Ta, T. C., Faris, M. A., & Coulman, B. E. (1989). Estimation of nitrogen fixation and transfer from alfalfa to associated grasses in mixed swards under field conditions. *Plant and Soil*, 114(2), 249–255. https://doi.org/10.1007/BF02220805
- Carlton, A. J., Cameron, K. C., Di, H. J., Edwards, G. R., & Clough, T. J. (2019). Nitrate leaching losses are lower from ryegrass/white clover forages containing plantain than from ryegrass/white clover forages under different irrigation. *New Zealand Journal of Agricultural Research*, 62(2), 150–172. https://doi.org/10.1080/00288233. 2018.1461659
- Cavers, P. B., Bassett, I. J., & Crompton, C. W. (1980). The biology of Canadian weeds.: 47. Plantago lanceolata L. Canadian Journal of Plant Science, 60(4), 1269–1282. https://doi.org/10.4141/cjps80-180
- Cherney, J. H., Smith, S. R., Sheaffer, C. C., & Cherney, D. J. R. (2020). Nutritive value and yield of reduced-lignin alfalfa cultivars in monoculture and in binary mixtures with perennial grass. *Agronomy Journal*, 112(1), 352–367. https://doi.org/10.1002/agj2.20045
- Darrow, K., & Deane Bowers, M. (1997). Phenological and population variation in iridoid glycosides of *Plantago lanceolata* (Plantaginaceae). *Biochemical Systematics and Ecology*, 25(1), 1–11. https:// doi.org/10.1016/S0305-1978(96)00090-7
- Dembek, R., Łyszczarz, R., & Zimmer-Grajewska, M. (2015). Plantago lanceolata L. as a component of permanent and renewed grasslands. Acta Scientiarum Polonorum— Agricultura, 13(4), 19–30. http://www.acta.media.pl/pl/full/1/ 2014/000010201400013000040001900030.pdf
- Deutsches Institut für Normung e.V. (DIN). (2005–2006). Bodenbeschaffenheit—Bestimmung von mineralischem Stickstoff (Nitrat und Ammonium) in Bodenprofilen (Nmin-Laborverfahren) (DIN 19746:2005-06). Beuth Verlag GmbH. https://www.beuth.de/de/norm/din-19746/77106268

- deWit, C. T. (1960). On competition (66.8th ed.). Verslagen Landbouwkundige Onderzoekingen. https://library.wur.nl/WebQuery/ wurpubs/fulltext/187113
- Dietz, M., Machill, S., Hoffmann, H. C., & Schmidtke, K. (2013). Inhibitory effects of Plantago lanceolata L. on soil N mineralization. *Plant and Soil*, 368(1–2), 445–458. https://doi.org/10.1007/s11104-012-1524-9
- Dollete, D., Lumactud, R. A., Carlyle, C. N., Szczyglowski, K., Hill, B., & Thilakarathna, M. S. (2024). Effect of drought stress on symbiotic nitrogen fixation, soil nitrogen availability and soil microbial diversity in forage legumes. *Plant and Soil*, 495(1–2), 445–467. https://doi.org/ 10.1007/s11104-023-06348-1
- Dreymann, S., Loges, R., & Taube, F. (2005). Schnittgutabfuhr oder Gründüngung? Auswirkung der Kleegras-Nutzung auf Nitrat im Sickerwasser und Folgefrüchte. In *Ende der Nische, Beiträge zur 8.* Wissenschaftstagung Ökologischer Landbau, Kasse University press GmbH. https://orgprints.dk/id/eprint/3700/1/3700.pdf
- DüV. (2017). Verordnung über die Anwendung von Düngemitteln, Bodenhilfsstoffen, Kultursubstraten und Pflanzenhilfsmitteln nach den Grundsätzen der guten fachlichen Praxis beim Düngen2 (Düngeverordnung—DüV). https://www.gesetze-im-internet.de/d_v_ 2017/BJNR130510017.html
- Ellenberg, H., & Leuschner, C. (2010). Vegetation Mitteleuropas mit den Alpen: In ökologischer, dynamischer und historischer Sicht; 203 Tabellen (6., vollst. neu bearb. und stark erw. Aufl.). UTB Botanik, Ökologie, Agrar- und Forstwissenschaften.
- European Medicines Agency. (2011). Assessment report on Plantago lanceolata L., folium. European Medicines Agency. https://www.ema.europa.eu/en/documents/herbal-report/finalassessment-report-plantago-lanceolata-l-folium_en.pdf
- European Union. (1991). Council Directive 91/676/EEC of 12 December 1991 concerning the protection of waters against pollution caused by nitrates from agricultural sources. https://eur-lex.europa.eu/legalcontent/EN/TXT/HTML/?uri=CELEX:31991L0676
- Fernandez, A., Sheaffer, C., Tautges, N., Putnam, D., & Hunter, M. (2019). Alfalfa, wildlife & the environment (2nd ed.). National Alfalfa and Forage Alliance. https://www.alfalfa.org/ pdf/alfalfaenvironment2.pdf
- Flynn, E. S., Moore, K. J., Singer, J. W., & Lamkey, K. R. (2013). Evaluation of grass and legume species as perennial ground covers in corn production. *Crop Science*, 53(2), 611–620. https://doi.org/10.2135/ cropsci2011.06.0306
- Frick, R., Suter, D., & Hirschi, H.-U. (2019). Versuche mit Wiesenschwingel: Zwei neue Sorten empfohlen. Agrarforschung Schweiz, 10(7–8), 276–281.
- Fustec, J., Lesuffleur, F., Mahieu, S., & Cliquet, J.-B. (2010). Nitrogen rhizodeposition of legumes. A review. Agronomy for Sustainable Development, 30(1), 57–66. https://doi.org/10.1051/agro/2009003
- Gardiner, C. A., Clough, T. J., Cameron, K. C., Di, H. J., Edwards, G. R., & De Klein, C. A. M. (2018). Potential inhibition of urine patch nitrous oxide emissions by Plantago lanceolata and its metabolite aucubin. *New Zealand Journal of Agricultural Research*, 61(4), 495–503. https://doi.org/10.1080/00288233.2017.1411953
- Ghimire, K., Gupta, S., Geng, S., Chen, S., Boe, A., & Wu, Y. (2021). Identification of physiological and morphological traits governing high water use efficiency in alfalfa. *Journal of Agronomy and Crop Science*, 207(4), 644–653. https://doi.org/10.1111/jac.12483

- Grey, T. L., Eason, K. M., Wells, L., & Basinger, N. T. (2019). Effects of temperature on seed germination of plantago lanceolata and management in *Carya illinoinensis* production. *Plants*, 8(9), 308. https:// doi.org/10.3390/plants8090308
- Hansen, B., Kristensen, E. S., Grant, R., Høgh-Jensen, H., Simmelsgaard, S. E., & Olesen, J. E. (2000). Nitrogen leaching from conventional versus organic farming systems—A systems modelling approach. *European Journal of Agronomy*, 13(1), 65–82. https://doi.org/10.1016/S1161-0301(00)00060-5
- Hartmann, S. (2013). BQSM -FM 2 = Luzernegras für mittlere Standorte: Baverische *Oualitätssaatgutmischungen* Feldfutterbau. für den Bayerisches Staatsministerium fiir Ernährung, Landwirtschaft, Forsten und Tourismus (StMELF). https://www.lfl.bayern.de/ipz/gruenland/024122/index.php#:~:text= Empfohlene%20Mischung&text=Auf%20Grund%20seines% 20Eigenschaftsprofils%20ist,Mischung%20also%20etwas%20zus% C3%A4tzliche%20Nutzungselastizit%C3%A4t
- Heichel, G. H., Barnes, D. K., Vance, C. P., & Henjum, K. I. (1984). N₂ fixation, and N and dry matter partitioning during a 4-year alfalfa stand. *Crop Science*, 24(4), 811–815. https://doi.org/10.2135/ cropsci1984.00111831002400040043xa
- Heichel, G. H., Vance, C. P., Barnes, D. K., & Henjum, K. I. (1985). Dinitrogen fixation, and n and dry matter distribution during 4 year stands of birdsfoot trefoil and red clover. *Crop Science*, 25(1), 101– 105. https://doi.org/10.2135/cropsci1985.0011183i002500010026x
- Heß, J. (1989). Kleegrasumbruch im Organischen Landbau: Stickstoffdynamik im Fruchtfolgeglied Kleegras—Kleegras-Weizen-Roggen [Doctoral dissertation, University of Bonn].
- Judson, H. G., Fraser, P. M., & Peterson, M. E. (2019). Nitrification inhibition by urine from cattle consuming *Plantago lanceolata*. *Journal* of New Zealand Grasslands, 81, 111–116. https://doi.org/10.33584/ jnzg.2019.81.413
- Kivelitz, H. (2020). Ackerfutter für die Frühjahrssaat. Landwirtschaftskammer Nordrhein-Westfalen. https://www.landwirtschaftskammer. de/riswick/versuche/pflanzenbau/futterbau/veroeffentlichungen/ ackerfutterbau-gruenland.htm
- Kayser, M., Müller, J., & Isselstein, J. (2010). Nitrogen management in organic farming: Comparison of crop rotation residual effects on yields, N leaching and soil conditions. *Nutrient Cycling in Agroecosystems*, 87(1), 21–31. https://doi.org/10.1007/s10705-009-9309-0
- Kelner, D. J., Vessey, J. K., & Entz, M. H. (1997). The nitrogen dynamics of 1-, 2- and 3-year stands of alfalfa in a cropping system. Agriculture, Ecosystems & Environment, 64(1), 1–10. https://doi.org/10. 1016/S0167-8809(97)00019-4
- Kirwan, L., Lüscher, A., Sebastià, M. T., Finn, J. A., Collins, R. P., Porqueddu, C., Helgadottir, A., Baadshaug, O. H., Brophy, C., Coran, C., Dalmannsdóttir, S., Delgado, I., Elgersma, A., Fothergill, M., Frankow-Lindberg, B. E., Golinski, P., Grieu, P., Gustavsson, A. M., Höglind, M., ... Connolly, J. (2007). Evenness drives consistent diversity effects in intensive grassland systems across 28 European sites. *Journal of Ecology*, 95(3), 530–539. https://doi.org/10.1111/j. 1365-2745.2007.01225.x
- Kling, C., & Bruckner, A. (2024). Etablierung von Luzerne mit Gemengepartnern unter heterogenen Bodenbedingungen in Brandenburg. In Landwirtschaft und Ernährung—Transformation macht nur gemeinsam Sinn: 17. Wissenschaftstagung Ökologischer Landbau. Justus-Liebig-Universität Göttingen. https://orgprints.org/id/eprint/ 53815/1/Beitrag_161_final_a.pdf

- Knapp, S., Gunst, L., M\u00e4der, P., Ghiasi, S., & Mayer, J. (2023). Organic cropping systems maintain yields but have lower yield levels and yield stability than conventional systems—Results from the DOK trial in Switzerland. *Field Crops Research*, 302, 109072. https://doi.org/10. 1016/j.fcr.2023.109072
- Koch, H.-J. (1997). Einfluss unterschiedlicher Brachebegrünung auf das Risiko der Nitratauswaschung. *Pflanzenbauwissenschaften*, 4(1), 154–163. https://www.openagrar.de/servlets/MCRFileNodeServlet/ openagrar_derivate_00021510/PBW-97-4-2.pdf
- Köhler, W., Schachtel, G., & Voleske, P. (2012). Biostatistik: Eine Einführung für Biologen und Agrarwissenschaftler (5., aktualisierte und erw. Aufl.). Springer. https://doi.org/10.1007/978-3-642-29271-2
- Krachunova, T., Scholz, M., Bellingrath-Kimura, S. D., & Schmidtke, K. (2023). Ridge cultivation for the adaption of fodder maize (*Zea mays* L.) to suboptimal conditions of low mountain ranges in organic farming in Central Europe. *Agriculture*, 13(3), 650. https://doi.org/10. 3390/agriculture13030650
- Landwirtschaftskammer, N. R. W. (2025). Ackerfutterbau. Rategeber Pflanzenbau und Pflanzenschutz 2025. Landwirtschaftskammer Nordrhein-Westfalen. https://www.landwirtschaftskammer.de/ landwirtschaft/ackerbau/zwischenfruechte/ackerfutterbau-pdf.pdf
- LfL. (2024). Integrierter Pflanzenschutz: Leitunkräuter in Getreide. https://www.lfl.bayern.de/mam/cms07/publikationen/daten/ merkblaetter/leitunkr%C3%A4uter_lfl-merkblatt.pdf
- LfL. (2025). Wiesenschwingel—Festuca pratensis Huds. (engl.: meadow fescue). https://www.lfl.bayern.de/ipz/gruenland/022445/index.php
- LfULG. (2020). LUIS—Landwirtschaft- und Umweltinformationssystem für Geodaten [Computer software]. Sächsisches Landesamt für Umwelt, Landwirtschaft und Geologie. Dresden, Germany. https:// luis.sachsen.de/boden/bk50.html
- Liu, X., Tahir, M., Li, C., Chen, C., Xin, Y., Zhang, G., Cheng, M., & Yan, Y. (2022). Mixture of alfalfa, orchardgrass, and tall fescue produces greater biomass yield in Southwest China. *Agronomy*, *12*(10), 2425. https://doi.org/10.3390/agronomy12102425
- Lorenz, F., & Erdle, K. (2018). Die Düngeverordnung Umsetzen: DLG Merkblatt 426. Deutsche Landwirtschaftsgesellschaft (DLG), Fachzentrum Landwirtschaft. Frankfurt am Main, Germany. https://www.dlg.org/fileadmin/downloads/landwirtschaft/themen/ publikationen/merkblaetter/dlg-merkblatt_426.pdf
- Mangwe, M. C., Bryant, R. H., Beck, M. R., Fleming, A. E., & Gregorini, P. (2020). Grazed chicory, plantain or ryegrass–white clover alters milk yield and fatty acid composition of late-lactating dairy cows. *Animal Production Science*, 60(1), 107–113. https://doi.org/10.1071/ AN18537
- Marak, H. B., Biere, A., & van Damme, J. M. M. (2003). Fitness costs of chemical defense in *Plantago lanceolata* L.: Effects of nutrient and competition stress. *Evolution: International Journal of Organic Evolution*, 57(11), 2519–2530. https://doi.org/10.1111/j.0014-3820. 2003.tb01496.x
- Masoni, A., Mariotti, M., & Ercoli, L. (2015). Nitrate leaching from forage legume crops and residual effect on Italian ryegrass. *Agrochimica*, 59, 75–91. https://doi.org/10.12871/0021857201515
- Moghaddam, A., Raza, A., Vollmann, J., Ardakani, M. R., Wanek, W., Gollner, G., & Friedel, J. K. (2015). Biological nitrogen fixation and biomass production stability in alfalfa (*Medicago sativa* L.) genotypes under organic management conditions. *Biological Agriculture* & *Horticulture*, 31(3), 177–192. https://doi.org/10.1080/01448765. 2014.1001437

- Munzert, M. (2015). Landwirtschaftliche und gartenbauliche Versuche mit SAS. Springer. https://doi.org/10.1007/978-3-642-54506-1
- Navarrete, S., Rodriguez, M., Horne, D., Hanly, J., Hedley, M., & Kemp, P. (2022). Nitrogen excretion by dairy cows grazing plantain (*Plantago lanceolata*) based pastures during the lactating season. *Animals:* An Open Access Journal from MDPI, 12(4), 469. https://doi.org/10. 3390/ani12040469
- Notaris, C., Rasmussen, J., Sørensen, P., & Olesen, J. E. (2018). Nitrogen leaching: A crop rotation perspective on the effect of N surplus, field management and use of catch crops. *Agriculture, Ecosystems & Environment*, 255, 1–11. https://doi.org/10.1016/j.agee.2017.12.009
- Nußbaum, H. (2007). Klee und Kleegras erfolgreich silieren. In Jahresinhalte Landinfo 2007. Pflanzliche Erzeugung Heft 3/07 8. Landwirtschaftliches Zentrum für Rinderhaltung, Grünlandwirtschaft, Milchwirtschaft, Wild und Fischerei Baden-Württemberg. https://lel.landwirtschaft-bw.de/,Lde/Startseite/ Service_+Downloads/Jahresinhalte+Landinfo+2007#pflanze
- Pandey, A., Li, F., Askegaard, M., Rasmussen, I. A., & Olesen, J. E. (2018). Nitrogen balances in organic and conventional arable crop rotations and their relations to nitrogen yield and nitrate leaching losses. *Agriculture, Ecosystems & Environment, 265*, 350–362. https://doi.org/10.1016/j.agee.2018.05.032
- Peterson, M. E., Fraser, P. M., Curtin, D., van Klink, J. W., & Joyce, N. I. (2022). Soil nitrification inhibition by urine of sheep consuming plantain (*Plantago lanceolata*). *Biology and Fertility of Soils*, 58(3), 265–276. https://doi.org/10.1007/s00374-021-01573-1
- Pijlman, J., Berger, S. J., Lexmond, F., Bloem, J., van Groenigen, J. W., Visser, E. J. W., Erisman, J. W., & van Eekeren, N. (2020). Can the presence of plantain (*Plantago lanceolata* L.) improve nitrogen cycling of dairy grassland systems on peat soils? *New Zealand Journal of Agricultural Research*, 63(1), 106–122. https://doi.org/10.1080/ 00288233.2019.1698620
- Pol, M., Potterat, O., Tröber, F., Lewandowska, S., & Schmidtke, K. (2024). Rooting patterns and aucubin content of *Plantago lanceolata* varieties. *Research Square*. https://doi.org/10.21203/rs.3.rs-3748622/ v1
- Pol, M., Schmidtke, K., & Lewandowska, S. (2021). Plantago lanceolata—An overview of its agronomically and healing valuable features. *Open Agriculture*, 6(1), 479–488. https://doi.org/10.1515/ opag-2021-0035
- Pons, T. L., & van der Toorn, J. (1988). Establishment of *Plantago lance-olata* L. and *Plantago major* L. Among grass: I. Significance of light for germination. *Oecologia*, 75(3), 394–399. https://doi.org/10.1007/ BF00376942
- Prince, S., Anower, M. R., Motes, C. M., Hernandez, T. D., Liao, F., Putman, L., Mattson, R., Seethepalli, A., Shah, K., Komp, M., Mehta, P., York, L. M., Young, C., & Monteros, M. J. (2022). Intraspecific variation for leaf physiological and root morphological adaptation to drought stress in alfalfa (*Medicago sativa* L.). *Frontiers in Plant Science*, 13, 795011. https://doi.org/10.3389/fpls.2022.795011
- Rakotovololona, L., Beaudoin, N., Ronceux, A., Venet, E., & Mary, B. (2019). Driving factors of nitrate leaching in arable organic cropping systems in Northern France. *Agriculture, Ecosystems & Environment*, 272, 38–51. https://doi.org/10.1016/j.agee.2018.10.005
- Reza, M. M., Redoy, M. R. A., Rahman, M. A., Ety, S., Alim, M. A., Cheng, L., & Al-Mamun, M. (2021). Response of plantain (*Plantago lanceolata* L.) supplementation on nutritional, endo-parasitic, and endocrine status in lambs. *Tropical Animal Health and Production*, 53(1), Article 82. https://doi.org/10.1007/s11250-020-02514-0

- Rigby, D., & Cáceres, D. (2001). Organic farming and the sustainability of agricultural systems. *Agricultural Systems*, 68(1), 21–40. https:// doi.org/10.1016/S0308-521X(00)00060-3
- Rumball, W., Keogh, R. G., Lane, G. E., Miller, J. E., & Claydon, R. B. (1997). 'Grasslands Lancelot' plantain (*Plantago lanceolata* L.). New Zealand Journal of Agricultural Research, 40(3), 373–377. https:// doi.org/10.1080/00288233.1997.9513258
- Sagar, G. R., & Harper, J. L. (1960). Factors affecting the germination and early establishment of plantains (Plantago lanceolata, P.media and P.major). Department of Agriculture Botony, University College of N. Wales. https://www.cabdirect.org/cabdirect/abstract/ 19612300501
- Schmidtke, K. (2001). Umweltgerechter Anbau von Leguminosen-Entwicklung und Anwendung eines Verfahrens zur Quantifizierung der N-Flächenbilanz (Az. 07312): Abschlußbericht des Forschungsvorhabens, gefördert von der Deutschen Bundesstiftung Umwelt, Osnabrück: 2001-12. Forschungsbericht. Eine Publikation der Georg-August-Universität Göttingen. https://publications.goettingen-research-online.de/handle/2/129536
- Shapiro, S. S., & Wilk, M. B. (1965). An analysis of variance test for normality (complete samples). *Biometrika*, 52(3/4), 591–611. https:// doi.org/10.2307/2333709
- Sheaffer, C. C., Tanner, C. B., & Kirkham, M. B. (1988). Alfalfa water relations and irrigation. In A. A. Hanson, D. K. Barnes, R. R. Hill, G. H. Heichel, K. T. Leath, O. J. Hunt, G. C. Marten, M. B. Tesar, S. H. Mickelson, K. A. Holtgraver, C. W. Stuber, & D. E. Kissel (Eds.), *Alfalfa and alfalfa improvement* (Vol. 29, pp. 373–409). American Society of Agronomy Inc. https://doi.org/10.2134/agronmonogr29. c11
- Simon, P. L., De Klein, C. A. M., Worth, W., Rutherford, A. J., & Dieckow, J. (2019). The efficacy of Plantago lanceolata for mitigating nitrous oxide emissions from cattle urine patches. *The Science of the Total Environment*, 691, 430–441. https://doi.org/10.1016/j.scitotenv. 2019.07.141
- Singh, A., Afzal, T., Woodbury, B., Wortmann, C., & Iqbal, J. (2023). Alfalfa in rotation with annual crops reduced nitrate leaching potential. *Journal of Environmental Quality*, 52(4), 930–938. https://doi. org/10.1002/jeq2.20473
- Skinner, R. H., & Gustine, D. L. (2002). Freezing tolerance of chicory and narrow-leaf plantain. *Crop Science*, 42(6), 2038–2043. https:// doi.org/10.2135/cropsci2002.2038
- Sleugh, B., Moore, K. J., George, J. R., & Brummer, E. C. (2000). Binary legume–grass mixtures improve forage yield, quality, and seasonal distribution. *Agronomy Journal*, 92(1), 24–29. https://doi.org/ 10.2134/agronj2000.92124x
- Soba, D., Zhou, B., Arrese-Igor, C., Munné-Bosch, S., & Aranjuelo, I. (2019). Physiological, hormonal and metabolic responses of two alfalfa cultivars with contrasting responses to drought. *International Journal of Molecular Sciences*, 20(20), 5099. https://doi.org/10.3390/ ijms20205099
- Stewart, A. V. (1996). Plantain (*Plantago lanceolata*)—A potential pasture species. In *Proceedings of the New Zealand Grassland Association, Oamaru*. (Vol. 58, pp. 77–86). New Zealand Grassland Association. https://doi.org/10.33584/jnzg.1996.58.2221
- Suter, D., Briner, H., Mosimann, E., & Stevenin, L. (2004). Sortenversuche mit Wiesenschwingel und Rotschwingel. AgrarForschung, 11(7), 274–279. https://www.agrarforschungschweiz.ch/wp-content/ uploads/2019/12/2004_07_879.pdf

- Suter, D., Frick, R., & Hirschi, H.-U. (2011). Sortenprüfung Wiesenschwingel: Bewährungsprobe für alt und neu. Agrarforschung Schweiz, 2(6), 258–263. https://www.agrarforschungschweiz.ch/wpcontent/uploads/2019/12/2011_06_1672.pdf
- Ta, T. C., & Faris, M. A. (1987). Species variation in the fixation and transfer of nitrogen from legumes to associated grasses. *Plant and Soil*, 98(2), 265–274. https://doi.org/10.1007/BF02374830
- Tamura, Y., & Nishibe, S. (2002). Changes in the concentrations of bioactive compounds in plantain leaves. *Journal of Agricultural and Food Chemistry*, 50(9), 2514–2518. https://doi.org/10.1021/ jf011490x
- Thompson, K., & Grime, J. P. (1983). A Comparative study of germination responses to diurnally-fluctuating temperatures. *The Journal of Applied Ecology*, 20(1), 141–156. https://doi.org/10.2307/2403382
- Tukey, J. W. (1949). Comparing individual means in the analysis of variance. *Biometrics*, 5(2), 99–114. https://doi.org/10.2307/3001913
- VDLUFA. (2012a). Bestimmung von mineralischem Stickstoff (Nitrat und Ammonium) in Bodenprofilen (N_{min}-Labormethode) (A 6.1.4.1): Teil A: Probenahme und chemische Untersuchungen, 4 Aufl. 1991. In R. Bassler, R. Thun, & G. Hoffmann (Eds.), *Handbuch der landwirtschaftlichen Versuchs- und Untersuchungsmethodik: Bd. 1,1. Die Untersuchung von Böden* (4., neubearb. u. erw. Aufl.). VDLUFA-Verl. http://www.methodenbuch.de/index.php?option=com_content&view=article&id=7&Itemid=108&lang=de
- VDLUFA. (2012b). Bestimmung von Phosphor und Kalium im Calcium-Acetat-Lactat-Auszug (CAL) A 6.2.1.1: Methodenbuch I, 6. Teillfg. 2012. In R. Bassler, R. Thun, & G. Hoffmann (Eds.), *Handbuch der landwirtschaftlichen Versuchs- und Untersuchungsmethodik: Bd. 1,1. Die Untersuchung von Böden* (4., neubearb. u. erw. Aufl.). VDLUFA-Verl.
- VDLUFA. (2012c). Bodenuntersuchungen Bestimmung des pH-Wertes A5.1.1: Methodenbuch I, 7. Teillfg. 2016. In R. Bassler, R. Thun, & G. Hoffmann (Eds.), Handbuch der landwirtschaftlichen Versuchsund Untersuchungsmethodik: Bd. 1,1. Die Untersuchung von Böden (4., neubearb. u. erw. Aufl.). VDLUFA-Verl.

- Veronesi, F., Brummer, E. C., & Huyghe, C. (2010). Alfalfa. In B. Boller, U. K. Posselt, & F. Veronesi (Eds.), *Fodder crops and amenity grasses* (pp. 395–437). Springer. https://doi.org/10.1007/978-1-4419-0760-8_17
- Walinga, I., van der Lee, J. J., Houba, V. J. G., van Vark, W., & Novozamsky, I. (1995). Digestion in tubes with H₂SO₄-salicylic acid-H₂O₂ and selenium and determination of Ca, K, Mg, N, Na, P, Zn. In I. Walinga (Ed.), *Plant analysis manual* (pp. 7–45). Springer. https://doi.org/10.1007/978-94-011-0203-2
- Walley, F. L., Tomm, G. O., Matus, A., Slinkard, A. E., & van Kessel, C. (1996). Allocation and cycling of nitrogen in an alfalfa-bromegrass sward. *Agronomy Journal*, 88(5), 834–843. https://doi.org/10.2134/ agronj1996.00021962008800050025x
- Wery, J., Turc, O., & Salsac, L. (1986). Relationship between growth, nitrogen fixation and assimilation in a legume (*Medicago sativa* L.). *Plant and Soil*, 96(1), 17–29. https://doi.org/10.1007/BF02374992
- Willey, R. W., & Osiru, D. S. O. (1972). Studies on mixtures of maize and beans (*Phaseolus vulgaris*) with particular reference to plant population. *The Journal of Agricultural Science*, 79(3), 517–529. https://doi.org/10.1017/S0021859600025909

SUPPORTING INFORMATION

Additional supporting information can be found online in the Supporting Information section at the end of this article.

How to cite this article: Krachunova, T., Bellingrath-Kimura, S., & Schmidtke, K. (2025). Annual ribwort plantain and alfalfa mixtures enhance forage accumulation and reduce nitrate. *Agronomy Journal*, *117*, e70098. https://doi.org/10.1002/agj2.70098